Author's response to reviews

Title: Reconstruction of a missed posterior locked shoulder fracture-dislocation with bone graft and lesser tuberosity transfer: a case report

Authors:

Byron E Chalidis (byronchalidis@gmail.com)
Pericles P Papadopoulos (perpap@otenet.gr)
Christos G Dimitriou (chdimit@otenet.gr)

Version: 2 Date: 10 April 2008

Author's response to reviews:

Dear Sir,

The authors have tried to address all the issues that were pointed out from the Editorial Team. All the corrections were made according to the provided instructions.

We really apologise for any inconsistency and we’re looking forward to receiving your comments and reply.

Yours sincerely

Byron E. Chalidis MD
Pericles P. Papadopoulos MD
Christos G. Dimitriou MD